Author’s response to reviews

Title: Tanzanian lessons in using non-physician clinicians to scale up comprehensive emergency obstetric care in remote and rural areas

Authors:

Angelo S Nyamtema (nyamtema_angelo@yahoo.co.uk)
Senga K Pemba (pemba@healthtrainingifakara.org)
Godfrey Mbaruku (gmbaruku@ihi.or.tz)
Fulgence D Rutasha (dadi@unfpa.org)
Jos van Roosmalen (J.J.M.van_Roosmalen@lumc.nl)

Version: 4 Date: 31 October 2011

Author’s response to reviews:

Major revisions (we require the author to make these changes)

References: reference 21, Submitted/ unpublished. Articles that have not been accepted for publication should not be cited in the reference list, unless they are publicly available as a pre-print. Instead, please refer to them in the text in the following way: (Authors: Title of manuscript, submitted).
Response: The article is already published. We have updated the reference.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above).

Abbreviations: Please put this section into paragraph form.
Response: The section is edited accordingly

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
Response: Attempts have been made to reduce the white space around the image. If not satisfactory please assist.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Response: The authors have taken time to read the article and few typographical errors have been identified and corrected.